Author's response to reviews

Title: Prevalence of von Hippel-Lindau gene mutations in sporadic renal cell carcinoma: results from the Netherlands cohort study

Authors:

Kjeld P. van Houwelingen (K.vanHouwelingen@ncmls.kun.nl)
Boukje A.C. van Dijk (Boukje.vanDijk@epid.unimaas.nl)
Christina A. Hulsbergen-van de Kaa (C.Hulsbergen@pathol.azn.nl)
Leo J. Schouten (L.J.Schouten@epid.unimaas.nl)
Hanneke J.M. Gorissen (H.Gorissen@ncmls.kun.nl)
Jack A. Schalken (J.Schalken@ncmls.kun.nl)
Piet A. van den Brandt (P.A.vandenBrandt@epid.unimaas.nl)
Egbert Oosterwijk (E.Oosterwijk@ncmls.kun.nl)

Version: 4 Date: 31 May 2005

Author's response to reviews:

31 May 2005

Dear editorial team,

We would like to thank you for the news that our article entitled "Prevalence of Von Hippel-Lindau gene mutations in sporadic Renal Cell Carcinoma: results from the Netherlands Cohort Study" (MS ID: 106522680582079) is now, in principle, accepted for publication in BMC Cancer. We have made changes according to all formatting changes requested. We have also checked and corrected typing errors and made some small changes in the entire manuscript.

With best regards, on behalf of the authors,

Boukje van Dijk